Author’s response to reviews

Title: The influence of the rs1137101 genotypes of leptin receptor gene on the demographic and metabolic profile of normal Saudi females and those suffering from polycystic ovarian syndrome

Authors:

Maha Daghestani (mdaghestani@ksu.edu.sa)
Mazen Daghestani (mhdaghestani@uqu.edu.sa)
Mamoon Daghistani (mdaghist@hotmail.com)
Geir Bjørklund (bjorklund@conem.org)
Salvatore Chirumbolo (salvatore.chirumbolo@univr.it)
Arjumand Warsy (aswarsy@gmail.com; aswarsy@ksu.edu.sa)

Version: 1 Date: 16 Aug 2017

Author’s response to reviews:

Dear Sir,

The response to each of the comments is as follows:

Comment 1. Please change the heading ‘Aim’ to ‘Methods’ in the abstract.
Response: 'Aim' is changed to 'Methods'.

Comment 2. In the section ‘Ethics approval and consent to participate’ please include the specific name of the institutional review board that ethically approved this study.
Response: The name of the institutional Review board has been included.

Comment 3. In the section 'Funding', please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Response: The role of the Funding body is described.
Comment 4. In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/submissions/editorial-policies#standards+of+reporting), could you please ensure your manuscript adheres to STROBE guidelines for cross-sectional studies and you attach the STROBE checklist as an additional file -  (https://www.strobe-statement.org/index.php?id=strobe-home)

Response: We have made sure that the manuscript adheres to STROBE guidelines and the checklist is included.